%
% dropoff.tex
%

\documentclass[12pt]{article}
\pagestyle{empty}

\setlength{\oddsidemargin}{0in}
%\setlength{\evensidemargin}{.25in}
\setlength{\textwidth}{6.5in}
\setlength{\topmargin}{-1.25in}
%\setlength{\headheight}{0in}
%\setlength{\headsep}{0in}
\setlength{\textheight}{11in}
\setlength{\parindent}{0in}

\begin{document}

\vfill

\begin{center}
\bf\Large Project Based Math 112, Fall 2001\\\LARGE
\bf \Large Instructions for turning in \\
homework outside of class\\
\end{center}

{\bf Normal homework drop-off:} Usually homework will be collected in
class. If this is not possible for some reason, you can drop off your
homework in a folder in 310 Malott Hall, next to the reception desk.
The office is open from 8 am to 4:30 pm. You should have your name,
course \#, section \#, date and instructor's name on each sheet of
paper. You should place the homework in the appropriate course file
folder.\\

{\bf After hours drop-off:} If you need to drop off your homework
after hours when the department is closed, you may do so at the main
desk of the Math Library on the 4th floor of Malott during Library
hours. There is a box labeled "Math Homework After-hours Drop-off".
These will be collected each morning and filed in the regular drop-off
file folders in 310 Malott.\\

{\bf Please notify your instructor via email when you leave something
in the folder. The folder will not be checked regularly.}\\


\vfill
\vfill


\begin{center}
\bf\Large Project Based Math 112, Fall 2001\\\LARGE
\bf \Large Instructions for turning in \\
homework outside of class\\
\end{center}

{\bf Normal homework drop-off:} Usually homework will be collected in
class. If this is not possible for some reason, you can drop off your
homework in a folder in 310 Malott Hall, next to the reception desk.
The office is open from 8 am to 4:30 pm. You should have your name,
course \#, section \#, date and instructor's name on each sheet of
paper. You should place the homework in the appropriate course file
folder.\\

{\bf After hours drop-off:} If you need to drop off your homework
after hours when the department is closed, you may do so at the main
desk of the Math Library on the 4th floor of Malott during Library
hours. There is a box labeled "Math Homework After-hours Drop-off".
These will be collected each morning and filed in the regular drop-off
file folders in 310 Malott.\\

{\bf Please notify your instructor via email when you leave something
in the folder. The folder will not be checked regularly.}\\

\vfill

\end{document}





